Author's response to reviews

Title: The Critical Elements of Effective Academic-Practice Partnerships: A Framework Derived from a Case Study of the Launch Year of the Department of Veterans Affairs Nursing Academy

Authors:

Aram Dobalian (aram.dobalian@va.gov)
Candice Bowman (candybowman@gmail.com)
Tamar Wyte-Lake (tamar.wyte@va.gov)
Marjorie L Pearson (mpearson@rand.org)
Mary B Dougherty (mary.dougherty@va.gov)
Jack Needleman (needlema@ucla.edu)

Version: 2

Date: 31 January 2014

Author's response to reviews:

The title has been revised to indicate that this is a case study design. A title page has been added. Information on competing interests, authors' contributions, and acknowledgements has been added to the revised cover letter.